Author's response to reviews

Title: Cost of Stroke: A Controlled National Study Evaluating Societal Effects on Patients and Their Partners

Authors:

Poul J Jennum (poul.joergen.jennum@regionh.dk)
Helle K Iversen (helle.klinigenberg.iversen@regionh.dk)
Rikke Ibsen (rikke@i2minds.dk)
Jakob Kjellberg (Jaki@kora.dk)

Version: 8 Date: 19 September 2015

Author's response to reviews: see over
Author's response to reviews

Title: Cost of Stroke: A Controlled National Study Evaluating Societal Effects on Patients and Their Partners

Authors:

Poul J Jennum (poul.joergen.jennum@regionh.dk)
Helle K Iversen (helle.klinigenberg.iversen@regionh.dk)
Rikke Ibsen (rikke@i2minds.dk)
Jakob Kjellberg (Jak@kora.dk)

Version: 7 Date: 29 August 2015

Author's response to reviews:

I have include original and separate files for figures and deleted in manus.